Author’s response to reviews

Title: Functional Results and Photic Phenomena with new Extended-Depth-of-Focus Intraocular Lens

Authors:
Bert Giers (b.giers@gmx.de)
Ramin Khoramnia (ramin.khoramnia@med.uni-heidelberg.de)
Dorottya Varadi (dorottya.varadi@med.uni-heidelberg.de)
Hannah Wallek (hannah.wallek@med.uni-heidelberg.de)
Hyeck Soo Son (Son.hyecksoo@gmail.com)
Mary Attia (marie.safwat1@gmail.com)
Gerd Auffarth (auffarthg@aol.com)

Version: 2 Date: 11 Aug 2019

Author’s response to reviews:

August 11th, 2019

Gerd Auffarth, MD, PhD
David J. Apple International Laboratory for Ocular Pathology
University Eye Clinic Heidelberg
Heidelberg, Germany

Brock Madsen
BMC Ophthalmology

Re: Manuscript, “Functional Results and Photic Phenomena with new Extended-Depth-of-Focus Intraocular Lens”
Dear Mr. Madsen,

thank you for your e-Mail of August 8th, 2019, regarding our manuscript. We are pleased at the positive reception of our work. Below we provide a point-by-point response to your comments:

Editor Comments:

“Please include the date of trial registration with the Trial registration number at the end of the abstract. If the date of registration is after the date that the first participant was recruited, please also state ‘retrospectively registered’. (i.e. “Trial Registration: [number] [date], [retrospectively registered if applicable]”).”

- We now added the date of trial registration at the end of our abstract (Page 2, Lines 29-30).

“Please add a heading "Declarations" after your List of abbreviations and before the References section. Under this heading be sure to add the following sub-sections: …”

- We now added the heading “Declarations” after our List of abbreviations and also added the given sub-sections.

“In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.”

- We now added this information under the Funding section.

“In the ethics approval and consent to participate subsection of the Declarations please clarify whether consent was obtained in written or verbal form. If consent was obtained verbally please explain why this was sought over written informed consent and provide a statement as to whether your ethics committee explicitly approved this method.”

- We now added this information under the Ethics Approval and Consent to Participate section.
“At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.”

- Our manuscript is now a single, final, clean version without any tracked changes or highlights.

Thank you again for your assistance with our manuscript. Please let us know if further issues arise.

We look forward to hear from you.

Sincerely yours,

Gerd Auffarth, MD, PhD

Corresponding Author